Himmelfarb Health Sciences Library, The George Washington University Health Sciences Research Commons

Environmental and Occupational Health Faculty Publications

Environmental and Occupational Health

11-2016

Leukocyte Telomere Length in Relation to 17 Biomarkers of Cardiovascular Disease Risk: A Cross-Sectional Study of US Adults

David Rehkopf

Belinda L. Needham

Jue Lin

Elizabeth Blackburn

Ami R. Zota George Washington University

See next page for additional authors

Follow this and additional works at: http://hsrc.himmelfarb.gwu.edu/sphhs_enviro_facpubs Part of the <u>Cardiovascular Diseases Commons</u>, <u>Cell Anatomy Commons</u>, <u>Cell Biology</u> <u>Commons</u>, and the <u>Genetics and Genomics Commons</u>

APA Citation

Rehkopf, D., Needham, B. L., Lin, J., Blackburn, E., Zota, A. R., Wojcicki, J., & Epel, E. (2016). Leukocyte Telomere Length in Relation to 17 Biomarkers of Cardiovascular Disease Risk: A Cross-Sectional Study of US Adults. *PLoS Medicine*, 13 (11). http://dx.doi.org/10.1371/journal.pmed.1002188

This Journal Article is brought to you for free and open access by the Environmental and Occupational Health at Health Sciences Research Commons. It has been accepted for inclusion in Environmental and Occupational Health Faculty Publications by an authorized administrator of Health Sciences Research Commons. For more information, please contact hsrc@gwu.edu.

Authors

David Rehkopf, Belinda L. Needham, Jue Lin, Elizabeth Blackburn, Ami R. Zota, Janet Wojcicki, and Elissa Epel



Citation: Rehkopf DH, Needham BL, Lin J, Blackburn EH, Zota AR, Wojcicki JM, et al. (2016) Leukocyte Telomere Length in Relation to 17 Biomarkers of Cardiovascular Disease Risk: A Cross-Sectional Study of US Adults. PLoS Med 13 (11): e1002188. doi:10.1371/journal. pmed.1002188

Academic Editor: Kazem Rahimi, University of Oxford, UNITED KINGDOM

Received: July 22, 2016

Accepted: October 25, 2016

Published: November 29, 2016

Copyright: © 2016 Rehkopf et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All data are available from the National Center for Health Statistics at http://wwwn.cdc.gov/nchs/nhanes/search/ nhanes99_00.aspx and https://wwwn.cdc.gov/ Nchs/Nhanes/Search/Nhanes01_02.aspx. R code for the analyses presented in this manuscript are available at https://purl.stanford. edu/rm601wy2649.

Funding: This research was funded by grant R01AG033592 from the National Institute on Aging. DHR is supported by the National Institute **RESEARCH ARTICLE**

Leukocyte Telomere Length in Relation to 17 Biomarkers of Cardiovascular Disease Risk: A Cross-Sectional Study of US Adults

David H. Rehkopf¹*, Belinda L. Needham², Jue Lin³, Elizabeth H. Blackburn³, Ami R. Zota⁴, Janet M. Wojcicki⁵, Elissa S. Epel⁶

1 Department of Medicine, Stanford University, Stanford, California, United States of America, 2 Department of Epidemiology, University of Michigan, Ann Arbor, Michigan, United States of America, 3 Department of Biochemistry and Biophysics, University of California San Francisco, San Francisco, California, United States of America, 4 Department of Environmental and Occupational Health, George Washington University, Washington, DC, United States of America, 5 Department of Pediatrics, University of California San Francisco, San Francisco, California, United States of America, 6 Department of Psychiatry, University of California San Francisco, S

* drehkopf@stanford.edu

Abstract

Background

Leukocyte telomere length (LTL) is a putative biological marker of immune system age, and there are demonstrated associations between LTL and cardiovascular disease. This may be due in part to the relationship of LTL with other biomarkers associated with cardiovascular disease risk. However, the strength of associations between LTL and adiposity, metabolic, proinflammatory, and cardiovascular biomarkers has not been systematically evaluated in a United States nationally representative population.

Methods and Findings

We examined associations between LTL and 17 cardiovascular biomarkers, including lipoproteins, blood sugar, circulatory pressure, proinflammatory markers, kidney function, and adiposity measures, in adults ages 20 to 84 from the cross-sectional US nationally representative 1999–2002 National Health and Nutrition Examination Survey (NHANES) (n = 7,252), statistically adjusting for immune cell type distributions. We also examine whether these associations differed systematically by age, race/ethnicity, gender, education, and income. We found that a one unit difference in the following biomarkers were associated with kilobase pair differences in LTL: BMI -0.00478 (95% CI -0.00749-0.00206), waist circumference -0.00211 (95% CI -0.00325--0.000969), percentage of body fat -0.00516 (95% CI -0.00761-0.0027), high density lipoprotein (HDL) cholesterol 0.00179 (95% CI 0.000571-0.00301), triglycerides -0.000285 (95% CI -0.000555--0.0000158), pulse rate -0.00194 (95% CI -0.00317-0.000705), C-reactive protein -0.0363 (95% CI 0.0601--0.0124), cystatin C -0.0391 (95% CI -0.0772--0.00107). When using clinical cut-points we additionally found associations between LTL and insulin resistance -0.0412 (95% CI -0.0685-0.0139), systolic blood pressure 0.0455 (95% CI 0.00137-0.0897), and diastolic blood pressure -0.0674 (95% CI -0.126-0.00889). These associations were 10%-15% greater without

of Aging (K01AG047280). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: I have read the journal's policy and the authors of this manuscript have the following competing interests: JL is a cofounder and consultant to Telomere Diagnostic Inc. (formerly Telome Health).

Abbreviations: CDC, US Centers for Disease Control and Prevention; HDL, high-density lipoprotein; LDL, low-density lipoprotein; LTL, leukocyte telomere length; NHANES, National Health and Nutrition Examination Survey; SNP, single nucleotide polymorphism. controlling for leukocyte cell types. There were very few differences in the associations by age, race/ethnicity, gender, education, or income. Our findings are relevant to the relationships between these cardiovascular biomarkers in the general population but not to cardiovascular disease as a clinical outcome.

Conclusions

LTL is most strongly associated with adiposity, but is also associated with biomarkers across several physiological systems. LTL may thus be a predictor of cardiovascular disease through its association with multiple risk factors that are physiologically correlated with risk for development of cardiovascular disease. Our results are consistent with LTL being a biomarker of cardiovascular aging through established physiological mechanisms.

Author Summary

Why Was This Study Done?

- Leukocyte telomere length (LTL) is a biomarker of white blood cell division and is strongly associated with age.
- There are still mixed findings on how LTL is related to different types of mortality.
- Observational evidence and quasi-experimental studies suggest that there is an association between shorter LTL and cardiovascular disease mortality.
- It is unclear, however, whether or not LTL is completely independent of already known biological pathways of cardiovascular disease.

What Did the Researchers Do and Find?

- We examined US nationally representative data on participants ages 25 to 84 that had LTL measured along with 17 other cardiovascular risk biomarkers.
- We found associations between LTL and measures of adiposity (BMI, waist circumference, percentage of body fat), along with C-reactive protein, cystatin C, high-density lipoprotein (HDL) cholesterol, triglycerides, insulin resistance, systolic blood pressure, diastolic blood pressure, and pulse rate.
- We did not find that there were a meaningful number of differences in these relationships by age, race/ethnicity, or socioeconomic position, suggesting that the associations we find are generally consistent across different population groups.

What Do These Findings Mean?

- Our findings suggest that LTL is associated with cardiovascular risk factors across different physiological systems.
- The study findings do not have any implications for whether LTL is a cause of cardiovascular disease.

Introduction

Shorter leukocyte telomere length (LTL) is significantly associated with increased risk of cardiovascular disease, irrespective of adjustment for conventional risk factors [1]. A meta-analysis of 24 studies with 43,725 participants and 8,400 patients with cardiovascular disease estimated a relative risk of 1.54 (95% CI 1.30-1.83), comparing those with the longest to shortest third of LTL [1]. Although these finding are consistent with a role for LTL in cardiovascular disease etiology, the critical details of how LTL is related to known metabolic pro-inflammatory markers and cardiovascular risk factors in the pathway to cardiovascular disease are unclear [2]. It is thought that telomere length shortens in relationship to multiple types of biochemical stressors and thus may represent a cumulative index of known risk factors. Our study will contribute to answering this question using a large US nationally representative sample with a comprehensive number of biomarkers measured along with LTL. Examining the relationship of telomere length with multiple biomarkers concurrently will inform us of whether LTL might be a marker for one primary risk factor, such as inflammation, or of multiple risk factors. This analysis will thus help to better understand whether LTL represents a unique biomarker that is independent from other known risk factors for cardiovascular disease.

In previous research, LTL was associated with myocardial infarction and stroke among younger (\leq 73 years) individuals [3]; rate of LTL shortening was associated with cardiovascular disease among elderly men [4]; LTL was associated with congestive heart failure [5]; and LTL was associated with prospective cardiovascular disease events [6]. Single nucleotide polymorphisms (SNPs) that are associated with telomere length are also associated with cardiovascular disease, offering a complementary source of evidence consistent with the observational data [7]. There are several prior studies that have each examined cardiovascular risk biomarkers and their association with LTL. The most comprehensive analysis to date examined approximately 2,500 community-dwelling white individuals from Belgium aged 35–55 who did not have diagnosed cardiovascular disease [8]. They found very little correlation with traditional lipid biomarkers but did find associations with markers of inflammation, including C-reactive protein. Other studies have found mixed associations between diastolic blood pressure and LTL, with one study finding an inverse association [3] and another no association [4]. A prior study of a nationally representative sample of 1,000 individuals in Costa Rica found a statistically significant positive association with systolic blood pressure [9].

To our knowledge, this is the first study to comprehensively examine the relationships between cardiovascular risk factor biomarkers and LTL in a large, US nationally representative sample, allowing us to either confirm or refute the results of recent studies on telomere length in relation to specific metabolic, immune function, and cardiovascular biomarkers from smaller, isolated populations and to perform analyses controlling for multiple measures of socioeconomic position, clinical variables, and chronic disease as potentially confounding variables. In addition, because we have a nationally representative study that oversampled racial/ ethnic minority populations, we can also appropriately test for effect measure modification by age, race/ethnicity, gender, education, and income. This is critical, as a prior meta-analysis suggested that there may be substantial heterogeneity across demographic groups in the relationship between LTL and myocardial infarction but this study did not have enough data to robustly test this hypothesis [10]. A further contribution of our work is examining these associations among individuals of all ages, because an important potential contribution of LTL is using it as a biomarker of aging across the life course. Prior studies have focused primarily on relationships with cardiovascular biomarkers in the population aged 65 and older [11]. In addition, the use of a wide range of ages is important because it allows for analyses conditional on current chronic disease, which is much more common among older individuals.

Because LTL comes from a mix of blood types that have systematically different extents of differentiation depending on function and thus have different mean telomere lengths, previously demonstrated associations of LTL and health outcomes may be confounded by cell type composition [12]. We thus also examine associations statistically controlling for cell type composition. Finally, given the possibility that a linear model may not be the appropriate functional form for examining the association between LTL and continuous cardiovascular biomarkers, we present sensitivity analyses fit with penalized spline models and also examine associations with clinical cut-points of biomarkers.

Materials and Methods

Study Population

We pooled data from the 1999-2000 and 2001-2002 cycles of National Health and Nutrition Examination Survey (NHANES), a nationally representative survey and physical examination of the civilian, noninstitutionalized US population conducted by the US Centers for Disease Control and Prevention (CDC). All data that we used for the analyses presented here are available for public download (http://www.cdc.gov/nchs/nhanes/nhanes_questionnaires.htm). Between 1999 and 2002, DNA samples were stored from NHANES participants aged 20 and older to establish a national probability sample of genetic material for future research. Our analysis is limited to these years of data because of the availability of DNA for LTL assay during only these years. We excluded individuals aged 85 and older from our analysis because of mortality selection over this age and because most traditional risk factors for disease are not predictive of morbidity and mortality in this population. Of the 9,191 participants aged 20-84 in the sample who were eligible to provide DNA, 7,525 (82%) provided DNA, consented to its use in future research, and had a sufficient quality and quantity of DNA to estimate telomere length. Non-Hispanic blacks, women, and subjects older than 60 years of age were less likely to give consent for future genetic research [13]. Participants provided written consent for participation in this study, and all study protocols were approved by the Institutional Review Board at the CDC.

LTL Measurements

Aliquots of purified DNA were provided by the National Center for Health Statistics. DNA was isolated from whole blood using the Puregene (D-50K) kit protocol (Gentra Systems, Inc., Minneapolis, Minnesota) and stored at -80°C. To measure mean LTL, quantitative PCR assay was used to determine the relative ratio of telomere repeat copy number to single-copy gene copy number (T/S ratio), with the full details of this standard protocol described elsewhere [14,15]. Each sample was assayed three times on three different days. The samples were assayed on duplicate wells, resulting in six measurements per sample. Sample plates were assayed in groups of three plates, and no two plates were grouped together more than once. Each assay plate contained 96 control wells. Any assay runs with eight or more invalid control wells were considered a failed run and were excluded from further analysis (<1% of runs). The mean of the T/S ratio values was calculated, and the largest and the smallest T/S ratio values in the set were marked as potential outliers. Then the mean of the T/S ratio value was calculated without the potential outliers. If the absolute value of the log of the ratio between the recalculated mean (excluding the potential outliers) to the value of the potential outlier was greater than 0.4, then the value was marked as an outlier (98.7% of all samples contained no outliers). The LTL assay lab was blind to the sample characteristics. The inter-assay coefficient of variability for LTL

was 4%. The equation for conversion from T/S ratio to kb pairs used was 3,274 + 2,413 * (T/S) / 1,000 [16]. This conversion ratio is likely to differ between laboratories and even between assays within the same lab; thus, exact kb pair values we report should be used as an approximation of actual telomere length. Although this does not impact the internal validity of the analyses presented here, the absolute level of base pair length should not be used in direct comparison to other studies. The assays were conducted in the Blackburn Laboratory at the University of California, San Francisco. Due to the extreme right skew (high values) of a small number of observations that are likely to be assay errors and not biologically plausible, we excluded the top 1% of measures (n = 76) from all of our analyses.

Blood Sample Characteristics

NHANES 1999–2002 measured a number of characteristics of the blood samples from which DNA was extracted. These characteristics included the following: white blood cells (SI), lymphocytes (%), monocytes (%), neutrophils (%), eosinophils (%), and basophils (%). Overall white blood cell count may act as a confounder because it can capture whether infection is occurring, which may have resulted in proliferation of additional new cells within each cell type, which may have different telomere lengths, and which may not be captured by the five cell type composition measures. Fig 1 shows the correlation structure between these six factors. The only strong correlation is between neutrophils and lymphocytes. This suggests that most of these measures are capturing unique information about the blood component cell type composition.

Biomarker Outcomes Examined

NHANES 1999–2002 allows for an examination of 17 factors within six categories of biomarkers that are risk factors for cardiovascular disease: lipoproteins (high-density lipoprotein [HDL] cholesterol, low density lipoprotein [LDL] cholesterol, triglycerides), blood sugar (glucose, hemoglobin A1c, insulin resistance), circulatory pressure (systolic blood pressure, diastolic blood pressure, pulse rate), proinflammatory (C-reactive protein, fibrinogen), kidney function (cystatin C, glomerular filtration rate, urine albumin to creatinine ratio [ACR]), and adiposity measures (body mass index [BMI], waist circumference, estimated percentage of body fat). The details of the measurement of these factors have been described in the NHANES documentation [17]. We used values of serum creatinine to estimate the glomerular filtration rate using the reexpressed four-variable MDRD Study equation (glomerular filtration rate = $175 \times$ standardized (S_{cr})^{-1.154} × (age)^{-0.203}) [18] without equivalization for race/ethnicity and gender (we statistically controlled for race/ethnicity and gender in our analyses). The glomerular filtration rate is presented as mL/min/1.73 m². Insulin resistance was measured by HOMA-IR, which was calculated as the product of fasting glucose (mmol/L) and fasting insulin (μ U/ML) divided by 22.5. Urine albumin creatinine ratio was calculated as urine albumin divided by urinary creatinine and reported as mg/g. Unless otherwise indicated, biomarkers are reported in SI units. Fig 2 shows the correlation structure of these 17 measures, ordered by a hierarchical clustering algorithm. Although we see an expected level of correlation between measures of adiposity, including BMI and waist circumference as well as hemoglobin A1c and glucose, most other measures are not substantially correlated (>0.6). This suggests that it is appropriate to examine this wide range of biomarkers that capture generally unique aspects of physiology within several different systems. Because the health risks of levels of each of these biomarkers may in some cases be nonlinear, we also examined each as a clinical threshold for a high-risk level of the biomarker. We used the following levels as indicating an increased level of clinical risk based on ATP III guidelines and other relevant literature: HDL cholesterol





doi:10.1371/journal.pmed.1002188.g001

(<40 mg/dl) [19], LDL cholesterol (\geq 160 mg/dl) [19], triglycerides (\geq 200 mg/dl) [19], glucose (>100 mg/dl) [20], hemoglobin A1c (>5.4%) [21], HOMA-IR (\geq 2.5) [22], systolic blood pressure (\geq 140 mm Hg) [23], diastolic blood pressure (\geq 90 mm Hg) [23], pulse rate (\geq 83 beats per minute) [24,25], C-reactive protein (>3.0 mg/L) [26], fibrinogen (>4.0 g/mL) [27], cystatin C (>1.29 mg/L) [28], estimated glomerular filtration rate (<60 mL/min/1.73 m²) [29], urine ACR (>30 mg/g) [30], BMI (\geq 30 kg/m²) [31], waist circumference (>102 cm for men, >88 cm for women) [31], and estimated percentage of body fat (>25% for men, >30% for women) [32]. We also examine a summary measure of cardiometabolic risk—the metabolic syndrome, which is a sum of whether a participant is above the clinical cut-point for waist circumference, triglycerides, HDL cholesterol, systolic blood pressure, and glucose. Individuals with a score of 3 or more are considered to have the metabolic syndrome [33].

Demographic Variables

All demographic variables that we use in our analysis are derived from respondent self-report, including race/ethnicity (white, black, Mexican American, and other), marital status (married

PLOS MEDICINE







doi:10.1371/journal.pmed.1002188.g002

or living with partner versus not), place of birth (outside of the US or within the US), level of educational attainment (less than high school, high school, more than high school), and occupation (white collar and professional; white collar, semi-routine; blue collar, high skill; blue collar, semi-routine; not working). The occupation measure was based on the self-reported type of work that was done the longest, using a previously described classification [34]. Our measure of income was the NHANES calculated poverty–income ratio (PIR), or the ratio of household income to poverty threshold adjusted for family size and inflation and used as a continuous measure.

Health and Health Behaviors

All health and health behavior measures were derived from self-reported data. Respondents were asked separate questions on whether they had "ever been told by a doctor or other health professional that" they had high blood pressure or hypertension, had diabetes, or had congestive heart failure, coronary heart disease, angina, angina pectoris, a heart attack, or a stroke.

Participants were also asked if they had ever been told of having "weak or failing kidneys? Do not include kidney stones, bladder infections, or incontinence." Participants were asked whether they had smoked at least 100 cigarettes in their entire life and whether they currently smoke cigarettes, which we label as ever smoked and current smoker. Participants were also asked whether they had participated in "moderate activities that cause only light sweating or a slight to moderate increase in breathing or heart rate. Some examples are brisk walking, bicycling for pleasure, golf and dancing" over the past 30 days, as well as whether they had participated in "vigorous activities for at least ten minutes that cause heavy sweating or large increases in breathing or heart rate? Some examples are running, lap swimming, aerobics classes, or fast bicycling" at least once over the last 30 days.

Missing Data

Of the 7,525 individuals in the analytic sample, there were the following numbers of individuals missing demographic, health behavior, and chronic disease variables: education (1), income (660), occupation (3,072), marital status (358), current smoker (14), vigorous physical activity (6), hypertension (75), and diabetes (3). We also imputed missing data for cardiovascular biomarkers that were missing the following number of observations: HDL cholesterol (6), LDL cholesterol (4,217), triglycerides (4,084), glucose (4,086), insulin (3,831), hemoglobin A1c (5), systolic blood pressure (241), diastolic blood pressure (241), pulse pressure (206), C-reactive protein (1), fibrinogen (2,783), cystatin C (3,711), glomerular filtration rate (4,086), urine albumin to creatinine ratio (100), BMI (197), waist circumference (224), and percentage of body fat (4,672). Note that samples were intentionally smaller for LDL cholesterol, triglycerides, glucose, and insulin because we examine these only in the subsample that fasted overnight, and special survey sample weights for this subsample are used. The estimated body fat from bioimpedance spectroscopy has a larger number of missing observations because this was only performed on participants up to the age of 49 years old. Cystatin C was measured only on participants aged 60 and above and on a 25% random sample of participants aged 12-59 years old. Fibrinogen was assayed only on individuals aged 40 and older. In order to avoid the bias caused by complete case analysis, we imputed missing variables for the demographic control variables and cardiovascular disease biomarkers using a nonparametric missing value imputation with Random Forest, implemented with the R package missForest [35] fit with 100 trees. This approach has several advantages for our particular study as compared to multiple imputation methods for missing data, including that it can be used with specialized survey functions and nonlinear models. This method of imputing missing data has been shown to compare favorably, with the missForest algorithm having the smallest prediction difference between the imputed and non-missing data sets [36]. The large number of missing observations for occupation mean that there will be some residual confounding to measurement error in the imputed data for occupation, but this approach is preferable to the larger degree of uncontrolled confounding that would exist if excluding occupation from the models [37]. Note that based on prior empirical work in the literature we did not impute missing values of our dependent variable, LTL, as this generally results in either no improvement in model accuracy or more biased results [38,39].

Statistical Analysis

Analyses were conducted in R using the survey package [40], with models fit using the svyglm() function that uses inverse probability weighting and design-based standard errors All analyses accounted for the clustered sampling design and the examination survey weights as required for appropriate analysis of this data [41]. For variables that were only measured for

individuals who participated in an overnight fast (LDL cholesterol, triglycerides, glucose, insulin), fasting examination survey weights were used.

We used multivariable regression models to assess the relationship between LTL and each biomarker. Consistent with most prior literature on biomarkers and telomere length, we examined telomere length as a linear variable.

For all analyses, we present a model 1 that shows the coefficient from a simple linear regression model for the cell sample characteristic or cardiovascular biomarker in relation to the dependent variable of LTL. In model 2, we fit a multiple linear regression model that included additional variables that were demonstrated by Needham et al. (2013) to be predictors of LTL in this study population: race/ethnicity (indicators for white, Mexican American, black, and other), education (indicators for less than high school and high school diploma, with some college or more as the omitted comparison category), income, occupation (white collar and professional; white collar, semi-routine; blue collar, high skill; blue collar, semi-routine; not working), married, foreign born, age (as continuous linear), and age-squared. The use of continuous linear age along with an age-squared term is important for analyses of LTL given the strength with which it is associated with age and the potential for nonlinearity in this association [42]. Model 3 additionally includes two measures of smoking and two measures of physical activity, as these may act as confounding factors of the relationship between LTL and cardiovascular biomarkers. We fit this as a separate model because we want to assess whether the associations observed between biomarkers and LTL are attenuated due to health-related behaviors that are associated with LTL. Model 4 additionally includes the six available cell type composition measures. We fit each of the above models both with continuous levels of the biomarkers and with the two category clinical risk threshold measures. We also performed sensitivity analysis, restricting the population to individuals that did not have chronic disease risk, in order to test whether our results for the general population also apply to a health population. In particular, these robustness checks of results allow us to address the possible limitation that the presence of chronic disease may cause associations between LTL and cardiovascular disease risk biomarkers.

We a priori specified that we would test for effect measure modification between the continuous measures of the biomarkers and LTL by gender, age (20–44 and 65–84), income (PIR), education (less than high school), and race/ethnicity (black and Mexican American) and report the magnitude of differences observed and the 95% CIs when interaction terms did not include the null of no association.

Finally, given that there are unclear priors on the best functional form to model the relationship between each of these biomarkers and LTL, we conducted sensitivity analysis using a penalized spline model that allows there to be a nonlinear relationship between dependent and independent variables. These models were fit using a generalized additive mixed model that fits a number of splines across the data then uses a penalized fitting algorithm to fit a number of slopes that best fit the data. We specified a maximum of four degrees of freedom for the spline basis, with the best fitting model fit by generalized cross validation [43].

Results

Sample Characteristics

Table 1 shows the weighted distribution of population demographic characteristics for the full NHANES 1999–2002 population aged 20–84, the population with measured telomere length, as well as the characteristics for the population with measured telomere length for which missing values were imputed. For the full NHANES population, the distribution of these factors reflects their distribution in the US noninstitutionalized population consistent with the

Table 1. Sample characteristics, ages 20–84, NHANES 1999–2002.

	Full NHANES (<i>n</i> = 9,191)	With Telomere Length ($n = 7,252$)	Final Imputed (<i>n</i> = 7,252)
	Proportion/mean	Proportion/mean	Proportion/mean
Age (mean)	45.5	45.7	45.7
Female	0.52	0.511	0.511
Race/ethnicity			
Black	0.108	0.093	0.093
White	0.707	0.727	0.727
Mexican American	0.071	0.071	0.071
Other race/ethnicity	0.113	0.109	0.109
Education			
Less than high school	0.216	0.212	0.212
High school	0.256	0.26	0.26
Some college	0.526	0.527	0.527
Poverty: Income ratio (mean)	2.98	3.02	3.00
Occupation			
White collar—high	0.241	0.247	0.412
Blue Collar—high	0.112	0.115	0.115
White collar—low	0.223	0.222	0.175
Blue collar—low	0.392	0.384	0.263
No work	0.033	0.032	0.029
Foreign born	0.154	0.147	0.148
Married	0.664	0.660	0.664
Smoked (ever)	0.497	0.503	0.503
Smoker (current)	0.247	0.248	0.248
Physical activity (moderate)	0.472	0.484	0.484
Physical activity (vigorous)	0.351	0.358	0.358
Kidney disease	0.021	0.021	0.021
Hypertension	0.254	0.255	0.255
Diabetes	0.066	0.066	0.066
Cardiovascular disease	0.08	0.08	0.08

doi:10.1371/journal.pmed.1002188.t001

sampling design of the NHANES study. The population with measured telomere length differs primarily because of who consented to the future use of their DNA. Although the populations are very similar, there are about 1% fewer black participants in the sample with telomere length, as well as fewer foreign-born individuals. The distributions by education, income, and occupation are similar. The sample distributions for the final imputed sample only differ notably for occupational classification due to the large number of missing responses for this question. Fig 3 shows the distribution of LTL.

LTL and Constituent Cell Types

Table 2 shows coefficients of association between cell types comprising the blood sample from which DNA was extracted and the LTL assayed from the leukocytes. The model-based coefficients in the table can be interpreted as the difference in 1 kb pair of LTL associated with a one unit change in the cell type. For example, the coefficient of 0.0272 for Basophils in model 1 means that a difference in 1 kb of telomere length is associated with 2.72% difference in Basophils. Model 1 presents the unadjusted correlation coefficients. Model 2 includes social and demographic factors. Model 3 additionally controls health behaviors. In the unadjusted model



doi:10.1371/journal.pmed.1002188.g003

Table 2. Association of LTL with concentration of white blood cells and constituent cell types, ages 20 to 84, NHANES 1999–2002.

	Model 1—unadjusted				demographic	adjusted	Model 3—demographic + health- related behaviors			
	coef	95% CI		coef	95% CI		coef	95%	o Cl	
White blood cells (SI)	-0.00984	-0.0219,	0.00224	-0.0161**	-0.0264,	-0.00576	-0.0152*	-0.0253,	-0.0051	
Lymphocytes (%)	0.00288*	0.0000759,	0.00568	0.0000529	-0.00229,	0.0024	-0.000106	-0.00241,	0.0022	
Monocytes (%)	-0.0126*	-0.0217,	-0.00358	-0.00136	-0.00985,	0.00713	-0.00197	-0.0105,	0.0066	
Neutrophils (%)	-0.00109	-0.00355,	0.00136	0.000169	-0.0019,	0.00224	0.000354	-0.00168,	0.00239	
Eosinophils (%)	-0.00989**	-0.016,	-0.00376	-0.00433	-0.0103,	0.00167	-0.00468	-0.0107,	0.0013	
Basophils (%)	0.0272	-0.0264,	0.0807	0.0519	-0.000992,	0.105	0.0493	-0.00445,	0.103	

Sample size for all analyses is *n* = 7,252. The dependent variable in all models is LTL (in kbp). Each line shows coefficients and 95% CIs from separate statistical models. Model 1 does not adjust for any covariables. Model 2 adjusts for the following covariables: race/ethnicity (white, Mexican American, black, and other), gender, foreign birthplace, education (less than high school, high school diploma, more than high school), class of work (white collar high, blue collar high, white collar low, blue collar low, no work), income, marital status (married or living with partner), age (as continuous), and age-squared. Model 3 additionally controls for ever smoked, current smoker, moderate physical activity, and vigorous physical activity.

**p* < 0.05

***p* < 0.01

doi:10.1371/journal.pmed.1002188.t002

1, monocytes and eosinophils were negatively associated with LTL, and lymphocytes were positively associated, with 95% CIs that did not include 0. After controlling for demographic factors, the concentration of white blood cells is significantly associated with LTL. The results from model 3 are similar, for which a higher level of white blood cells is associated with lower LTL. We also examined this model in a restricted population of individuals aged 25 to 84 who did not have kidney disease, hypertension, diabetes, or cardiovascular disease, and associations were either similarly close to 0 or the magnitude of the association did not differ by more than 10% (S1 Table).

We also examined interactions with gender, race/ethnicity (black and Mexican American), education (less than high school diploma), income (continuous), and age (25–44 and 65–84)— a total of 42 interactions (S2 Table). We find a more negative association for monocytes for black participants (interaction coefficient = -0.0202; 95% CI -0.0336--0.00684), a more positive association of eosinophils with higher income (interaction coefficient = 0.00469; 95% CI -0.00149-0.00789), and both lower lymphocytes (interaction coefficient = -0.00348; 95% CI -0.00657--0.000391) and higher neutrophils (interaction coefficient = 0.00290; 95% CI 0.000418-0.00539) among participants age 65 and older. We thus found four interactions for which the null of 0 was not included in the confidence interval when 2–3 would be expected by chance alone.

LTL and Biomarkers

Table 3 presents coefficients of the association between biomarkers and LTL in the categories of lipoproteins, blood sugar, circulatory pressure, proinflammatory, kidney function, and adiposity measures. There were associations with LTL for 13 of the 17 biomarkers examined in the unadjusted models (model 1) for which the null association of 0 was not within the estimated 95% CIs, with eight of these associations remaining after controlling for demographic factors, health behaviors, and cell type (model 4). These factors were HDL cholesterol, triglycerides, pulse rate, C-reactive protein, cystatin C, BMI, waist circumference, and percentage of body fat. There was no association with the summary measure of metabolic syndrome after controlling for demographic characteristics, nor when additionally controlling for healthrelated behaviors. There was generally between a 10%-15% reduction in the magnitude of the coefficient after controlling for cell type (comparing models 2 and 3 to model 4), consistent with the fact that there may be some confounding of the associations we observed in models 2 and 3 by cell type. Fig 4 shows a dot plot of coefficients and 95% CIs comparing the magnitudes of association with LTL across each of the 17 biomarkers and the summary measure of the metabolic syndrome. The coefficients are from the same model as model 4, but the biomarkers are standardized (by dividing by their standard deviation), so the magnitude of the associations with LTL can be directly compared to each other. The scale of interpretation is how much of a standard deviation of the biomarker is associated with a 1 kb pair difference in LTL. The strongest negative correlations were for percentage of body fat, BMI, C-reactive protein, and waist circumference, followed by pulse. The strongest positive correlation was for HDL, which was of a similar magnitude of association as pulse rate. Note that HDL is the only biomarker we examined where higher levels are associated with lower cardiovascular disease risk [44].

We also examined the full model 4 in a restricted population of individuals aged 25 to 84 who did not have kidney disease, hypertension, diabetes, or cardiovascular disease (<u>S3 Table</u>). Coefficients of association were similar between the full population and the restricted population with the exception of cystatin C, for which coefficient estimates did not have overlapping 95% CIs. In the full population the association with LTL was -0.0391 (95% CI -0.0772–-0.00701),

1999–2002.
, NHANES
ges 20 to 84
omarkers, a
ntinuous bi
f LTL on co
Regression o
able 3. I

	Mode	el 1-unadjus	ted	Model 2-d	emographic	adjusted	Model 3-d	emographic	+ health-	Model 4-0	lemographic -	- health-
	coef	956	°C	coef	62%	° CI	coef	95%	Ū	coef	62%	CI
Lipoproteins							-					
HDL cholesterol	0.000941	-0.000445,	0.00233	0.00212**	0.000906,	0.00334	0.00206**	0.000839,	0.00328	0.00179*	0.000571,	0.00301
LDL cholesterol	-0.000963*	-0.00186,	-0.0000644	0.000271	-0.000491,	0.00103	0.0000858	-0.000601,	0.000773	0.000238	-0.000547,	0.00102
Triglycerides	-0.000946***	-0.00135,	-0.000546	-0.00035*	-0.000642,	-0.0000577	-0.000192	-0.000421,	0.0000383	-0.000285*	-0.000555,	-0.0000158
Blood sugar												
Glucose	-0.00281***	-0.00374,	-0.00187	-0.0000894	-0.000721,	0.000542	-0.000266	-0.000835,	0.000304	-0.00000737	-0.000604,	0.000589
Insulin resistance	-0.0135**	-0.0229,	-0.00421	-0.00483	-0.0117,	0.00203	-0.00309	-0.0063,	0.000114	-0.00324	-0.00934,	0.00286
HbA1c	-0.0765***	-0.0951,	-0.0578	-0.014	-0.0312,	0.00326	-0.0138	-0.0312,	0.00365	-0.00951	-0.0274,	0.00843
Circulatory pressure												
Systolic blood pressure	-0.00553***	-0.00659,	-0.00448	0.000703	-0.000243,	0.00165	0.000732	-0.000219,	0.00168	0.000815	-0.000158,	0.00179
Diastolic blood oressure	0.0000221	-0.00137,	0.00142	-0.0000155	-0.00139,	0.00136	-0.0000635	-0.00143,	0.0013	-0.000138	-0.00154,	0.00126
Pulse rate	-0.000143	-0.00151,	0.00123	-0.00248**	-0.00375,	-0.00121	-0.00226**	-0.00352,	-0.00101	-0.00194*	-0.00317,	-0.000705
Pro-inflammatory												
C-reactive protein	-0.0699***	-0.101,	-0.039	-0.0453**	-0.0693,	-0.0213	-0.043**	-0.0671,	-0.0188	-0.0363*	-0.0601,	-0.0124
Fibrinogin	-0.117***	-0.154,	-0.081	-0.0297	-0.0718,	0.0124	-0.0263	-0.0671,	0.0145	-0.0154	-0.0535,	0.0227
Kidney function												
Cystatin C	-0.273***	-0.368,	-0.178	-0.051*	-0.0942,	-0.00771	-0.0474*	-0.0878,	-0.00695	-0.0391*	-0.0772,	-0.00107
Glomerular filtration rate	0.00227***	0.0019,	0.00264	0.00000239	-0.000554,	0.000559	0.0000848	-0.000464,	0.000633	0.000158	-0.000402,	0.000719
Albumin: Creatinine	-0.0000444	-0.000107,	0.0000177	0.0000192	-0.0000392,	0.0000776	0.0000203	-0.0000375,	0.000078	0.0000221	-0.0000341,	0.0000782
Adiposity												
BMI	-0.00752***	-0.0107,	-0.00432	-0.00545**	-0.00808,	-0.00282	-0.00552**	-0.0083,	-0.00275	-0.00478*	-0.00749,	-0.00206
waist circumference	-0.00538***	-0.00676,	-0.00399	-0.00247**	-0.00362,	-0.00132	-0.00242**	-0.0036,	-0.00124	-0.00211*	-0.00325,	-0.000969
% body fat	-0.00334***	-0.00498,	-0.00171	-0.00609***	-0.00862,	-0.00355	-0.00575**	-0.00828,	-0.00322	-0.00516*	-0.00761,	-0.0027
Metabolic Syndrome	-0.0734***	-0.0931,	-0.0538	-0.0153	-0.0317,	0.0011	-0.013	-0.0284,	0.00238	-0.00985	-0.0244,	0.00469
The sample size for con	nponents that i	used the fast	ing weights w	as 3,407 (LDI	- cholesterol,	triglycerides,	glucose, and	l insulin resis	tance); sam	ole size for all	other measure	es was
7,252. The dependent v	ariable in all m	odels is LTL	(in kbp). Eacl	h line shows c	oefficients ar	nd 95% Cls fr	om separate	statistical mo	dels. Model	1 does not ad	ust for any co	variables.
Model 2 adjusts for the i	following covai	riables: race/	ethnicity (whi	te, Mexican A	merican, blac	k, and other)	, gender, fore	ign birthplace	e, education	(less than hig	h school, high	school
diploma, more than high	n school), class	s of work (wh	ite collar high	, blue collar hi	gh, white coll	ar low, blue c	ollar low, no v	vork), income	e, marital sta	tus (married o	r living with pa	artner), age
(as continuous), and ag	e-squared. Mc	odel 3 additio	nally controls	for ever smok	ed, current s	moker, mode	rate physical	activity, and	vigorous phy	sical activity.	Model 4 addit	onally
includes white blood ce	lls (SI), lymphc	ocytes (%), m	ionocytes (%)), basophils (%	6), eosinophil	s (%), and ne	utrophils (%)					

PLOS Medicine | DOI:10.1371/journal.pmed.1002188 November 29, 2016

p* < 0.05 *p* < 0.01 ****p* < 0.001.

HDL									
LDL				<u> </u>		-			
Trigylcerides				-					
Glucose									
Insulin resistance									
Svetolio BP						_			
Systolic BP						-			
									-
Pulse									
C-reactive Protein									
Fibrinogen									
Cycstatin C			·····-						
Glomelular Filtration Rate					_	-			
Albumin : Creatinine									
BMI		·····							
Waist									
Fat									
Metabolic Syndrome			<u>-</u>			_			
,									
	ا 0.06–	-0	.04	ا 0.02–	ا 0.00	ا 0.0	02 0.	04 0.0	06
Fig 4 Regression coefficients for	r relationsh	in hetween	hiomarker	e and ITI in	standard o	leviations of	hiomarker ner	kh of ITI age	s 20_84

Fig 4. Regression coefficients for relationship between biomarkers and LTL, in standard deviations of biomarker per kb of LTL, ages 20–84. NHANES 1999–2002.

doi:10.1371/journal.pmed.1002188.g004

whereas in the restricted healthy population the association with LTL was stronger, -0.24 (95% CI -0.39–-0.904).

We note that only a small number of interactions were observed based on our a priori specification to test for heterogeneity of association based on seven characteristics: female, black, Mexican American, less than a high school education, poverty income ratio, and ages 20 to 44 and ages 65 to 84. Based on these seven factors with 17 cardiovascular biomarkers, we tested 119 interactions (S4 Table). We note interactions for which the null of 0 for the interaction coefficient was not included in the estimated 95% CI. For triglycerides, there was a stronger association for women (interaction coefficient 0.000485; 95% CI 0.0000731–0.000897); for LDL cholesterol, there was a stronger relationship among black participants (interaction coefficient = 0.0020; 95% CI 0.000287–0.00376); for glomerular filtration rate, there was a stronger relationship among black participants (interaction coefficient = 0.000951, 95% CI 0.0000755– 0.00183); for percentage of body fat, there was a stronger relationship among black participants (interaction coefficient = 0.00697, 95% CI 0.00311-0.0108); for triglycerides, there was a stronger relationship among Mexican Americans (interaction coefficient = 0.00624, 95% CI 0.0000165–0.00123); for HbA1c, there was a stronger relationship among those with less than a high school diploma (interaction coefficient = 0.042, 95% CI 0.0158-0.0683); for percentage of body fat, there was a weaker relationship among those with lower levels of income (interaction coefficient = -0.00148, 95% CI -0.00232--0.000643); for diastolic blood pressure, there was a weaker relationship among participants aged 25-44 (interaction coefficient = -0.00288, 95% CI -0.0051--0.000652); and for glomerular filtration rate, there was a stronger relationship among participants aged 65 and above (interaction coefficient = 0.00107, 95% CI 0.0000132-0.00212). We found a total of nine interactions for which CIs did not include the null of no association, but, as we tested 119 interactions, we would expect six by chance alone. We therefore do not find substantial evidence for there being any systematic differences in associations in different groups of the population, but there is some evidence that there may be stronger associations between LTL and cardiovascular risk biomarkers among black participants.

We also repeated the primary analyses using dichotomous risk indicators of biomarkers, and results were generally similar (Table 4). Most of the associations with continuous measures were replicated in terms of magnitude, even though coefficients differed. Although associations with clinical thresholds of lipids were notably weaker than with the continuous measures, CIs for estimates were generally overlapping. The primary difference was that there was a stronger association with diastolic blood pressure and insulin resistance when using the clinical cut-point that was not observed for the continuous measure.

Table 5 considers a model that includes all of the biomarkers in the same multivariable model so as to examine whether predictors remain associated independently of other cardio-vascular disease biomarkers. Although interpretation of the coefficients should be done with caution because they are conditional on other biomarkers and the causal ordering of these biomarkers is not well characterized, this model provides an additional perspective on which cardiovascular biomarkers are the strongest relative predictors of LTL. In the model that includes no demographic controls, triglycerides, systolic blood pressure, diastolic blood pressure, cystatin C, glomerular filtration rate, BMI, waist circumference, and body fat percentage are associated with LTL. After controlling for demographic factors, only the coefficient for the association. However, these results should be interpreted in conjunction with the other findings, as domains that have more correlated measures, for example measures of adiposity, will be more attenuated by controlling for all of these measures in a single model.

Finally, we also examine the potential for nonlinear relationships between levels of cardiovascular biomarkers and LTL (Fig 5). In general, relationships do not deviate strongly from linearity, albeit with a few exceptions. With glucose, HbA1c, glomerular filtration rate, and systolic blood pressure, there is a weakening of the association above 6.5 kb pairs. Thus, even though there is a nonlinear relationship, there tends to be a linear association across the bulk of the data (between 5 and 6.5 kb pairs as shown in Fig 3), with nonlinearity existing primarily above 6.5 kb pairs. The exception to this is for diastolic blood pressure, for which there is an inverted U-shaped relationship.

Discussion

The purpose of our study was to examine whether LTL is independent of other known biological risk markers of cardiovascular disease. This is an important question given associations of

	Model 1—unadjusted		Model 2—demographic adjusted			Model 3—demographic + clinical adjusted			Model 4—demographic + clinical + cell type adjusted			
	coef	95%	% CI	coef	95	% CI	coef	95%	% CI	coef	95%	% CI
Lipoproteins												
HDL cholesterol	-0.035	-0.0772,	0.00711	-0.043*	-0.0784,	-0.0076	-0.0388*	-0.0736,	-0.00393	-0.0311	-0.0655,	0.00339
LDL cholesterol	0.0131	-0.0641,	0.0903	0.0638	-0.0087,	0.136	0.0594	-0.014,	0.133	0.0581	-0.0156,	0.132
Triglycerides	-0.138***	-0.199,	-0.0762	-0.062*	-0.118,	-0.00628	-0.0576*	-0.111,	-0.00422	-0.0504	-0.102,	0.00152
Blood sugar												
Glucose	-0.171***	-0.222,	-0.12	0.012	-0.0339,	0.058	0.0154	-0.0291,	0.0598	0.0228	-0.02,	0.0655
Insulin resistance	-0.138***	-0.166,	-0.111	-0.055***	-0.0807,	-0.0292	-0.0527**	-0.0783,	-0.0271	-0.0412*	-0.0685,	-0.0139
HbA1c	-0.175***	-0.212,	-0.138	-0.012	-0.0581,	0.0342	-0.00734	-0.0546,	0.04	0.00645	-0.0423,	0.0552
Circulatory pressu	ire											
Systolic blood pressure	-0.22***	-0.27,	-0.17	0.0442*	0.00103,	0.0873	0.0449*	0.00149,	0.0884	0.0455*	0.00137,	0.0897
Diastolic blood pressure	-0.0872**	-0.141,	-0.0331	-0.067*	-0.125,	-0.00918	-0.066*	-0.123,	-0.00858	-0.0674*	-0.126,	-0.00889
Pulse rate	0.0123	-0.0249,	0.0495	-0.0307	-0.0641,	0.0028	-0.0258	-0.058,	0.00649	-0.0161	-0.0482,	0.0161
Immune function												
C-reactive protein	-0.189*	-0.363,	-0.015	-0.13	-0.264,	0.00361	-0.119	-0.254,	0.0155	-0.0931	-0.225,	0.0391
Fibrinogin	-0.185***	-0.235,	-0.135	-0.0257	-0.0811,	0.0297	-0.0225	-0.0762,	0.0312	-0.00777	-0.0575,	0.0419
Kidney function												
Cystatin C	-0.352***	-0.421,	-0.283	-0.0345	-0.0989,	0.0298	-0.0334	-0.0965,	0.0297	-0.0279	-0.0909,	0.0351
Glomerular filtration rate	-0.386***	-0.506,	-0.267	-0.0744	-0.167,	0.0177	-0.0695	-0.16,	0.021	-0.0608	-0.155,	0.033
Albumin: Creatinine	-0.158***	-0.223,	-0.0924	-0.0312	-0.0912,	0.0289	-0.0285	-0.0884,	0.0313	-0.0235	-0.0819,	0.0349
Adiposity												
BMI	-0.0838***	-0.118,	-0.0498	-0.0617**	-0.0936,	-0.0298	-0.0616**	-0.0937,	-0.0294	-0.0526*	-0.0827,	-0.0225
waist circumference	-0.129***	-0.167,	-0.0921	-0.0635**	-0.0987,	-0.0283	-0.0617**	-0.0962,	-0.0271	-0.0509*	-0.0843,	-0.0175
% body fat	-0.188***	-0.231,	-0.145	-0.0489*	-0.0975,	-0.000342	-0.0437	-0.0921,	0.00468	-0.0357	-0.0848,	0.0134
Metabolic syndrome	-0.0734***	-0.0931,	-0.0538	-0.0153	-0.0317,	0.0011	-0.013	-0.0284,	0.00238	-0.00985	-0.0244,	0.00469

Table 4. Regression of LTL on clinical cut-points of biomarkers, ages 20–84, NHANES 1999–2002.

The sample size for components that used the fasting weights was 3,407 (LDL cholesterol, triglycerides, glucose, and insulin resistance); sample size for all other measures was 7,252. Model 4 adjusts for the following covariables: race/ethnicity (white, Mexican American, black, and other), gender, foreign birthplace, education (less than high school, high school diploma, more than high school), class of work (white collar high, blue collar high, white collar low, blue collar low, no work), income, marital status (married or living with partner), age (as continuous), age-squared, white blood cells (SI), lymphocytes (%), monocytes (%), basophils (%), eosinophils (%), neutrophils (%), ever smoked, current smoker, moderate physical activity, and vigorous physical activity. *p < 0.05

**p<0.01

***p<0.001.

doi:10.1371/journal.pmed.1002188.t004

LTL with cardiovascular disease mortality. The significance of this question is whether across the life course LTL is independent of already well-documented biological mechanisms of cardiovascular disease risk. Our findings suggest that, among a nationally representative US population, there are moderate correlations between LTL and important cardiovascular risk factors, even after accounting for age, social and demographic factors, health behaviors, and white blood cell types. We found that controlling for cell type reduced magnitudes of association by between 10% and 20%.

	Mode	l 1—unadjus	sted	Model 2—demographic adjuste			
	coef	95% CI		coef	95% CI		
Lipoproteins							
HDL cholesterol	-0.0284	-0.0676,	0.0108	0.0158	-0.0224,	0.054	
LDL cholesterol	-0.00285	-0.024,	0.0183	0.00938	-0.00903,	0.0278	
Triglycerides	-0.0505*	-0.0902,	-0.0107	-0.0194	-0.055,	0.0163	
Blood sugar							
Glucose	-0.013	-0.0721,	0.0462	0.0271	-0.0305,	0.0846	
Insulin resistance	0.0414*	-0.0142,	0.097	0.0101	-0.039,	0.0593	
HbA1c	-0.0277	-0.0913,	0.0359	-0.0172	-0.0796,	0.0452	
Circulatory pressure	·						
Systolic blood pressure	-0.0789*	-0.119,	-0.0385	0.00514	-0.0291,	0.0394	
Diastolic blood pressure	0.0336*	0.00886,	0.0584	-0.0143	-0.0417,	0.013	
Pulse rate	-0.015	-0.0384,	0.00827	-0.0314	-0.0557,	-0.00706	
Proinflammatory							
C-reactive protein	-0.0138	-0.0663,	0.0387	-0.0286	-0.0781,	0.021	
Fibrinogin	-0.0158	-0.0627,	0.031	0.015	-0.0313,	0.0612	
Kidney function							
Cystatin C	-0.0292**	-0.0483,	-0.0101	-0.0136	-0.0377,	0.0105	
Glomerular filtration rate	0.123	0.0858,	0.16	-0.00206	-0.0465,	0.0424	
Albumin: Creatinine	0.0434	-0.0159,	0.103	0.0119	-0.0451,	0.069	
Adiposity							
BMI	0.0705	0.00938,	0.132	-0.0417	-0.108,	0.0241	
waist circumference	-0.0809*	-0.147,	-0.0147	0.0306	-0.0359,	0.0971	
% body fat	-0.0415	-0.0763,	-0.0067	-0.00753	-0.0477,	0.0327	

Table 5.	Multivariable rec	aression of LTL	on all biomarkers	. age 20-84	NHANES	1999-2002
1 4 5 10 01	in and tan labit to	9.000.01.01 ET		,		1000 2002

The effective sample size for each of the models was 3,407 because of the use of fasting sample weights. Model 1 includes only the 17 biomarkers shown in the table. Model 2 additionally adjusts for the following covariables: race/ethnicity (white, Mexican American, black, and other), gender, foreign birthplace, education (less than high school, high school diploma, more than high school), class of work (white collar high, blue collar high, white collar low, blue collar low, no work), income, marital status (married or living with partner), age (as continuous), and age-squared.

doi:10.1371/journal.pmed.1002188.t005

We found association with LTL for two measure of lipoprotein (HDL cholesterol, triglycerides), one measure of blood sugar (insulin resistance), three measures of circulatory pressure (systolic blood pressure, diastolic blood pressure, and pulse rate), one proinflammatory marker (C-reactive protein), one measure of kidney function (cystatin C), and all three measures of adiposity (BMI, waist circumference, percentage of body fat). Among adiposity measures, percentage of body fat had the strongest association in independent models. LTL was not associated with a measure of overall cardiovascular risk—the metabolic syndrome because of its unique association with particular cardiovascular biomarkers. This study also allowed us to examine whether these relationships differed by age, gender, race/ethnicity, education, and income. This question is important because relationships between LTL and mortality have been found to differ by some of these domains [45]. In examining interactions between LTL and the 17 biomarkers, we found only slightly more interactions than would be expected by chance alone, suggesting that the relationships we present are fairly consistent across demographic groups. The strengths of this study are that we for the first time examine telomere length with biomarkers that are risk factors for morbidity and mortality in a multi-ethnic nationally representative sample of the US population. We also examined these associations across a broad age range of the population, from 25 to 84, based on interest in how LTL acts as a marker of biological aging across the life course. In addition, we examine this in a general population sample controlling for a greater number of potentially confounding factors as compared to other studies. We also are able to examine a wide range of cardiovascular risk biomarkers given the detailed measurements available in the NHANES data. No studies we are aware of have been able to examine this many biomarkers in a comparative manner nor to control for white cell type frequencies. Our findings should be considered generalizable to the US noninstitutionalized population from the ages of 25 to 84.

The study has several limitations. Most importantly, our results do not contribute to an understanding of the directionality of any associations because of the cross-sectional nature of the data. Causal inference from our findings is not possible. This refers to both the direction of potential causation between cardiovascular risk markers and LTL as well as the potential confounding of the biomarkers themselves. For example, it may be a dietary factor, rather than adiposity, that is the direct cause of changes in LTL. However, the goal of our study is to determine whether LTL is largely independent of other cardiovascular disease risk markers, and, thus, causal inference is not our intention. Randomized trials and quasi-experimental methods are more suited to the separate and also important question of how LTL may act as a cause or marker for cardiovascular disease risk. Secondly, although not a limitation per se, our results must be interpreted specifically with respect to LTL, because telomere length is cell type specific. Moreover, although we found that controlling for broad measures of cell type reduced magnitudes of association, we were not able to examine specific types of T and B cells where different telomere lengths have been demonstrated [15]. All of our conclusions about which physiological systems LTL is related to are limited by the commonly used clinical measures of those systems. In addition, we were not able to perform a mediation analysis for cardiovascular mortality due to the weak relationship between LTL and relatively short-term (mean 9.5 years) cardiovascular mortality follow-up available from NHANES [45]. In addition, although we use standard clinical cut-points for our analysis, we do not take into account that these may differ for best differentiating risk in population subgroups. Finally, our baseline data come from 1999-2002 and not from more recent data collection. Several facts mitigate this potential limitation. First, the actual telomere length assays were done in 2011, using more recent methodological approaches, and, thus, the assays themselves were done recently on stored DNA. Secondly, most of the population aged 25-84 that were part of the collection of data in 1999-2002 are still alive, so inference is still about the current US population. We are also not aware of any data suggesting that the relationship between LTL and cardiovascular biomarkers has changed since 1999-2002.

The results from our study can be compared to other studies that have previously examined similar factors and LTL. As mentioned in the introduction, the largest prior analysis found similar findings with respect to adiposity related measures and inflammation [8], with our study confirming these findings in a larger multi-ethnic US population. The most commonly examined cardiovascular risk factor with LTL has been BMI, and a recent meta-analysis summarized the results of five prior studies that controlled for covariables in regression analyses, finding an overall adjusted regression coefficient of -0.008 kBP per BMI unit [46] from 4,183 individuals. Our single study of a larger nationally representative population found a weaker association of -0.0047 kBP per BMI unit, which may in part be due to our ability to control for a greater number of potentially confounding factors.

We also found inverse associations with an inflammatory marker (C-reactive protein) in contrast with prior, more limited studies. A prior study in a sample of individuals over the age



Fig 5. Generalized additive mixed models of the association between biomarkers and LTL (in kb pairs), ages 20–84, NHANES 1999–2002.

doi:10.1371/journal.pmed.1002188.g005

of 70 did not find an association with C-reactive protein [47]. The differences in our findings are, however, consistent with genetic instrumental variables findings showing SNPs for C-reactive protein are associated with LTL [48]. In comparison to prior work [49], we did not find associations with measures of blood sugar and systolic blood pressure. Our findings for systolic blood pressure were also different from findings in a Costa Rican population, in which higher systolic blood pressure was found to be associated with longer telomere length [9].

Our findings can also be compared to those from an environmental wide association study between LTL and 461 environmental, behavioral, and clinical variables using the same NHANES data [50]. This study found 22 out of 461 associations, with false discovery rate of <5%. Comparable findings similar to those in our study include C-reactive protein, trunk fat (a measure of adiposity), and pulse rate. Our approach of targeting 17 cardiovascular disease biomarkers with a hypothesis-driven approach gave us more power to detect associations than in this prior study, which is the most likely explanation for our findings with additional adiposity measures, diastolic blood pressure, and HDL cholesterol. In addition, we controlled for white blood cell type, examined clinical cut-points of biomarkers, and examined nonlinearity of associations. As shown in our study, we missed the association between LTL and diastolic blood pressure when using a continuous measure rather than a clinical cut-point. Our findings were also similar to this prior study in finding little evidence for effect measure modification by gender and race/ethnicity.

Our study has implications for understanding how LTL may be relevant as a biomarker of aging. LTL is most associated with adiposity, immune function, diastolic blood pressure, pulse, and HDL cholesterol. This gives suggestions to the underlying role for LTL as a marker for coronary heart disease risk through oxidative stress-related pathways resulting in immune cell turnover and subsequent shorter LTL [47], consistent with the links between inflammation, adiposity, and oxidative stress [51]. Furthermore, our evidence is consistent with LTL as related to different aspects of biological dysregulation. Given the range of biomarkers measured in this large sample, it enabled us to test whether LTL truly reflects cumulative dysregulation across systems. The findings suggest that telomere length is indeed related to biomarkers of multiple regulatory systems that indicate risk for cardiovascular disease and possibly other diseases as well. There were relations across systems has been measured as a summary variable of unhealthy values across systems called allostatic load. Here, it appears LTL may serve as a cellular-based indicator of systemic allostatic load.

However, there were not relations across all systems; specifically, there were weaker or nonexistent relationships with blood sugar. With respect to this finding, prior studies have found differences in LTL between those with and without diabetes [52], which may be a more extreme comparison than looking at whether there is an association in a representative sample of the population. We do find consistent results in our basic model that does not control for demographic confounding variables, but this association is eliminated after controlling for a large set of potential confounding factors, which may also explain differences in results. A meta-analysis identifying an association between LTL and diabetes did not examine what potential confounding factors were controlled for [53]. In addition, studies showing greater mortality for those with shorter telomere length within diabetic populations [54] are more informative about the role of telomere length in the progression of disease rather than in the general population. With respect to kidney function, many prior studies reporting an association with LTL were in clinical samples, not in the general population [55]. Other work showing an association with kidney function and aging has been using telomere length from kidney tissue, not leukocytes [56]. Thus, although blood sugar and kidney function clearly contribute to cardiovascular risk, their dysregulation in the general population does not seem to be captured

by LTL, with the exception of some of our models showing a weak association with cystatin C. Again, however, it is important to consider that we used only established clinical risk factors for cardiovascular disease to capture the association of LTL with different physiological systems. Work taking a metabolomics approach has identified more specific biological components associated with LTL [57].

Future studies should continue to investigate the relationship between known cardiovascular biomarkers and LTL in more dynamic ways, including changes over time in both biomarkers and LTL. Most critically, in order to understand the clinical relevance for cardiovascular disease, longer duration follow-up studies that allow for mediation analysis with clinical endpoints must be designed to identify where there may be value added for the assay of LTL beyond traditional cardiovascular biomarkers. Finally, more complete measures of proinflammatory response should be examined to more specifically investigate how LTL is a marker of this system, ideally on cells that have been sorted by type [15].

Supporting Information

S1 STROBE Checklist. (DOC)

S1 Table. Association of LTL with concentration of white blood cells and constituent cell types, ages 20 to 84, population without chronic disease, NHANES 1999–2002. (DOCX)

S2 Table. Interaction terms for association of LTL with concentration of white blood cells and constituent cell types, ages 20 to 84, NHANES 1999-2002. (DOCX)

S3 Table. Regression of LTL on continuous biomarkers, ages 20 to 84, population without chronic disease, NHANES 1999-2002. (DOCX)

S4 Table. Interaction terms for association of LTL on continuous biomarkers, ages 20 to 84, NHANES 1999-2002. (DOCX)

S5 Table. Multivariable regression of LTL on all biomarkers, ages 20-84, population without chronic disease, NHANES 1999-2002. (DOCX)

Author Contributions

Conceptualization: DHR BLN JL EHB ESE.

Data curation: DHR BLN.

Formal analysis: DHR BLN.

Funding acquisition: DHR ESE.

Investigation: DHR BLN JL EHB ARZ JMW ESE.

Methodology: DHR BLN ARZ JMW ESE.

Resources: IL EHB.

Software: DHR.

Validation: DHR JL.

Visualization: DHR.

Writing - original draft: DHR.

Writing - review & editing: DHR BLN JL EHB ARZ JMW ESE.

References

- 1. Haycock PC, Heydon EE, Kaptoge S, Butterworth AS, Thompson A, Willeit P. Leucocyte telomere length and risk of cardiovascular disease: systematic review and meta-analysis. 2014.
- Sanders JL, Newman AB. Telomere length in epidemiology: a biomarker of aging, age-related disease, both, or neither? Epidemiologic reviews. 2013; 35(1):112–31.
- Fitzpatrick AL, Kronmal RA, Gardner JP, Psaty BM, Jenny NS, Tracy RP, et al. Leukocyte telomere length and cardiovascular disease in the cardiovascular health study. American Journal of Epidemiology. 2007; 165(1):14–21. Epub 2006/10/18. doi: 10.1093/aje/kwj346 PMID: 17043079.
- Epel ES, Merkin SS, Cawthon R, Blackburn EH, Adler NE, Pletcher MJ, et al. The rate of leukocyte telomere shortening predicts mortality from cardiovascular disease in elderly men. Aging (Albany NY). 2009; 1(1):81–8. Epub 2009/01/01. PMID: 20195384; PubMed Central PMCID: PMC2830080. doi: 10. 18632/aging.100007
- van der Harst P, van der Steege G, de Boer RA, Voors AA, Hall AS, Mulder MJ, et al. Telomere length of circulating leukocytes is decreased in patients with chronic heart failure. Journal of the American College of Cardiology. 2007; 49(13):1459–64. doi: 10.1016/j.jacc.2007.01.027 PMID: 17397675
- Willeit P, Willeit J, Brandstätter A, Ehrlenbach S, Mayr A, Gasperi A, et al. Cellular aging reflected by leukocyte telomere length predicts advanced atherosclerosis and cardiovascular disease risk. Arteriosclerosis, thrombosis, and vascular biology. 2010; 30(8):1649–56. doi: <u>10.1161/ATVBAHA.110.205492</u> PMID: 20508208
- Codd V, Nelson CP, Albrecht E, Mangino M, Deelen J, Buxton JL, et al. Identification of seven loci affecting mean telomere length and their association with disease. Nat Genet. 2013; 45(4):422–7, 7e1-2. doi: 10.1038/ng.2528 PMID: 23535734; PubMed Central PMCID: PMCPMC4006270.
- Bekaert S, De Meyer T, Rietzschel ER, De Buyzere ML, De Bacquer D, Langlois M, et al. Telomere length and cardiovascular risk factors in a middle-aged population free of overt cardiovascular disease. Aging Cell. 2007; 6(5):639–47. doi: 10.1111/j.1474-9726.2007.00321.x PMID: 17874998.
- DeFries T, Avendano M, Glymour MM. Level and change in cognitive test scores predict risk of first stroke. J Am Geriatr Soc. 2009; 57(3):499–505. Epub 2009/01/30. doi: 10.1111/j.1532-5415.2008. 02132.x PMID: 19175440.
- D'Mello MJ, Ross SA, Briel M, Anand SS, Gerstein H, Paré G. Association Between Shortened Leukocyte Telomere Length and Cardiometabolic Outcomes Systematic Review and Meta-Analysis. Circulation: Cardiovascular Genetics. 2015; 8(1):82–90.
- Guzzardi MA, Iozzo P, Salonen M, Kajantie E, Eriksson JG. Rate of telomere shortening and metabolic and cardiovascular risk factors: A longitudinal study in the 1934–44 Helsinki Birth Cohort Study. Annals of medicine. 2015; 47(6):499–505. doi: 10.3109/07853890.2015.1074718 PMID: 26339993
- Lin J, Cheon J, Brown R, Coccia M, Puterman E, Aschbacher K, et al. Systematic and Cell Type-Specific Telomere Length Changes in Subsets of Lymphocytes. Journal of immunology research. 2016;2016.
- McQuillan GM, Pan Q, Porter KS. Consent for genetic research in a general population: an update on the National Health and Nutrition Examination Survey experience. Genet Med. 2006; 8(6):354–60. doi: 10.109701.gim.0000223552.70393.08 PMID: 16778597.
- Cawthon RM. Telomere measurement by quantitative PCR. Nucleic Acids Res. 2002; 30(10):e47. Epub 2002/05/10. PMID: 12000852; PubMed Central PMCID: PMC115301.
- Lin J, Epel E, Cheon J, Kroenke C, Sinclair E, Bigos M, et al. Analyses and comparisons of telomerase activity and telomere length in human T and B cells: insights for epidemiology of telomere maintenance. Journal of immunological methods. 2010; 352(1–2):71–80. Epub 2009/10/20. doi: 10.1016/j.jim.2009. 09.012 PMID: 19837074; PubMed Central PMCID: PMC3280689.
- Farzaneh-Far R, Lin J, Epel E, Lapham K, Blackburn E, Whooley MA. Telomere length trajectory and its determinants in persons with coronary artery disease: longitudinal findings from the heart and soul study. PLoS ONE. 2010; 5(1):e8612. Epub 2010/01/15. doi: 10.1371/journal.pone.0008612 PMID: 20072607; PubMed Central PMCID: PMC2797633.

- Gunter EW, Lewis BG, Koncikowski SM. Laboratory Procedures Used for the Third National Health and Nutrition Examination Survey (NHANES III), 1988–1994. Hyattsville: National Center for Health Statistics, 1996.
- Levey AS, Coresh J, Greene T, Stevens LA, Zhang YL, Hendriksen S, et al. Using standardized serum creatinine values in the modification of diet in renal disease study equation for estimating glomerular filtration rate. Annals of internal medicine. 2006; 145(4):247–54. PMID: 16908915
- Grundy SM, Cleeman JI, Merz CNB, Brewer HB, Clark LT, Hunninghake DB, et al. Implications of recent clinical trials for the national cholesterol education program adult treatment panel III guidelines. Journal of the American College of Cardiology. 2004; 44(3):720–32. doi: 10.1016/j.jacc.2004.07.001 PMID: 15358046
- Seshasai SRK, Kaptoge S, Thompson A, Di Angelantonio E, Gao P, Sarwar N, et al. Diabetes mellitus, fasting glucose, and risk of cause-specific death. The New England journal of medicine. 2011; 364 (9):829. doi: 10.1056/NEJMoa1008862 PMID: 21366474
- **21.** Gerstein HC. Glycosylated hemoglobin: finally ready for prime time as a cardiovascular risk factor. Annals of internal medicine. 2004; 141(6):475–6. PMID: 15381522
- Kuk JL, Ardern CI. Are metabolically normal but obese individuals at lower risk for all-cause mortality? Diabetes care. 2009; 32(12):2297–9. doi: 10.2337/dc09-0574 PMID: 19729521
- Chobanian AV, Bakris GL, Black HR, Cushman WC, Green LA, Izzo JL, et al. Seventh report of the joint national committee on prevention, detection, evaluation, and treatment of high blood pressure. Hypertension. 2003; 42(6):1206–52. doi: 10.1161/01.HYP.0000107251.49515.c2 PMID: 14656957
- Fox K, Borer JS, Camm AJ, Danchin N, Ferrari R, Sendon JLL, et al. Resting heart rate in cardiovascular disease. Journal of the American College of Cardiology. 2007; 50(9):823–30. doi: 10.1016/j.jacc. 2007.04.079 PMID: 17719466
- Diaz A, Bourassa MG, Guertin M-C, Tardif J-C. Long-term prognostic value of resting heart rate in patients with suspected or proven coronary artery disease. European heart journal. 2005; 26(10):967– 74. doi: 10.1093/eurheartj/ehi190 PMID: 15774493
- Buckley DI, Fu R, Freeman M, Rogers K, Helfand M. C-reactive protein as a risk factor for coronary heart disease: a systematic review and meta-analyses for the US Preventive Services Task Force. Annals of internal medicine. 2009; 151(7):483–95. PMID: 19805771
- Ernst E, Resch KL. Fibrinogen as a cardiovascular risk factor: a meta-analysis and review of the literature. Annals of Internal Medicine. 1993; 118(12):956–63. PMID: 8489110
- Shlipak MG, Sarnak MJ, Katz R, Fried LF, Seliger SL, Newman AB, et al. Cystatin C and the risk of death and cardiovascular events among elderly persons. New England Journal of Medicine. 2005; 352 (20):2049–60. doi: 10.1056/NEJMoa043161 PMID: 15901858
- Consortium CKDP. Association of estimated glomerular filtration rate and albuminuria with all-cause and cardiovascular mortality in general population cohorts: a collaborative meta-analysis. The Lancet. 2010; 375(9731):2073–81.
- Levey AS, Eckardt KU, Tsukamoto Y, Levin A, Coresh J, Rossert J, et al. Definition and classification of chronic kidney disease: a position statement from Kidney Disease: Improving Global Outcomes (KDIGO). Kidney Int. 2005; 67(6):2089–100. doi: 10.1111/j.1523-1755.2005.00365.x PMID: 15882252.
- Janssen I, Katzmarzyk PT, Ross R. Body mass index, waist circumference, and health risk: evidence in support of current National Institutes of Health guidelines. Archives of internal medicine. 2002; 162 (18):2074–9. PMID: 12374515
- Williams DP, Going SB, Lohman TG, Harsha DW, Srinivasan SR, Webber LS, et al. Body fatness and risk for elevated blood pressure, total cholesterol, and serum lipoprotein ratios in children and adolescents. American journal of public health. 1992; 82(3):358–63. PMID: 1536350
- Loucks EB, Magnusson KT, Cook S, Rehkopf DH, Ford ES, Berkman LF. Socioeconomic position and the metabolic syndrome in early, middle, and late life: evidence from NHANES 1999–2002. Ann Epidemiol. 2007; 17(10):782–90. doi: 10.1016/j.annepidem.2007.05.003 PMID: 17697786.
- Rehkopf DH, Berkman LF, Coull B, Krieger N. The non-linear risk of mortality by income level in a healthy population: US National Health and Nutrition Examination Survey mortality follow-up cohort, 1988–2001. BMC Public Health. 2008; 8(1):383.
- Stekhoven DJ, Bühlmann P. MissForest—non-parametric missing value imputation for mixed-type data. Bioinformatics. 2012; 28(1):112–8. doi: 10.1093/bioinformatics/btr597 PMID: 22039212
- Waljee AK, Mukherjee A, Singal AG, Zhang Y, Warren J, Balis U, et al. Comparison of imputation methods for missing laboratory data in medicine. BMJ open. 2013; 3(8):e002847. doi: <u>10.1136/bmjopen-2013-002847</u> PMID: 23906948

- Braveman PA, Cubbin C, Egerter S, Chideya S, Marchi KS, Metzler M, et al. Socioeconomic status in health research: one size does not fit all. JAMA. 2005; 294(22):2879–88. Epub 2005/12/15. 294/22/ 2879 [pii] doi: 10.1001/jama.294.22.2879 PMID: 16352796.
- Von Hippel PT. Regression with missing Ys: An improved strategy for analyzing multiply imputed data. Sociological Methodology. 2007; 37(1):83–117.
- Sullivan TR, Salter AB, Ryan P, Lee KJ. Bias and Precision of the "Multiple Imputation, Then Deletion" Method for Dealing With Missing Outcome Data. American journal of epidemiology. 2015; 182(6):528– 34. doi: 10.1093/aje/kwv100 PMID: 26337075
- 40. Lumley T. Survey: analysis of complex survey samples. R package version 3.28–2. 2012.
- CDC. Analytic and Reporting Guidelines, The National Health and Nutrition Examination Survey Data Hyattsville: U.S. Department of Health and Human Services, Centers for Disease Control and Prevention; 2005 [cited 2007 October 28]. http://www.cdc.gov/nchs/data/nhanes/nhanes_03_04/nhanes_ analytic_guidelines_dec_2005.pdf.
- Rehkopf DH, Dow WH, Rosero-Bixby L, Lin J, Epel ES, Blackburn EH. Longer leukocyte telomere length in Costa Rica's Nicoya Peninsula: a population-based study. Exp Gerontol. 2013; 48(11):1266– 73. doi: 10.1016/j.exger.2013.08.005 PMID: 23988653; PubMed Central PMCID: PMCPMC3819141.
- **43.** Wood S. Generalized Additive Models: An Introduction with R. Carlin BP, Chatfield C, Tanner M, Zidek J, editors. Boca Raton: Chapman & Hall/CRC; 2006.
- 44. Yaari S, Even-Zohar S, Goldbourt U, Neufeld H. Associations of serum high density lipoprotein and total cholesterol with total, cardiovascular, and cancer mortality in a 7-year prospective study of 10 000 men. The Lancet. 1981; 317(8228):1011–5.
- 45. Needham BL, Rehkopf D, Adler N, Gregorich S, Lin J, Blackburn EH, et al. Leukocyte telomere length and mortality in the National Health and Nutrition Examination Survey, 1999–2002. Epidemiology. 2015; 26(4):528–35. doi: 10.1097/EDE.00000000000299 PMID: 26039272.
- Müezzinler A, Zaineddin A, Brenner H. Body mass index and leukocyte telomere length in adults: a systematic review and meta-analysis. Obesity Reviews. 2014; 15(3):192–201. doi: <u>10.1111/obr.12126</u> PMID: 24165286
- O'Donovan A, Pantell MS, Puterman E, Dhabhar FS, Blackburn EH, Yaffe K, et al. Cumulative inflammatory load is associated with short leukocyte telomere length in the Health, Aging and Body Composition Study. PLoS ONE. 2011; 6(5):e19687. doi: 10.1371/journal.pone.0019687 PMID: 21602933
- Rode L, Nordestgaard BG, Weischer M, Bojesen SE. Increased body mass index, elevated C-reactive protein, and short telomere length. The Journal of Clinical Endocrinology & Metabolism. 2014; 99(9): E1671–E5.
- 49. Demissie S, Levy D, Benjamin E, Cupples L, Gardner J, Herbert A, et al. Insulin resistance, oxidative stress, hypertension, and leukocyte telomere length in men from the Framingham Heart Study. Aging cell. 2006; 5(4):325–30. doi: 10.1111/j.1474-9726.2006.00224.x PMID: 16913878
- Patel CJ, Manrai AK, Corona E, Kohane IS. Systematic correlation of environmental exposure and physiological and self-reported behaviour factors with leukocyte telomere length. International journal of epidemiology. 2016:dyw043.
- Houben JM, Moonen HJ, van Schooten FJ, Hageman GJ. Telomere length assessment: biomarker of chronic oxidative stress? Free Radical Biology and Medicine. 2008; 44(3):235–46. doi: <u>10.1016/j.</u> freeradbiomed.2007.10.001 PMID: 18021748
- Testa R, Olivieri F, Sirolla C, Spazzafumo L, Rippo MR, Marra M, et al. Leukocyte telomere length is associated with complications of type 2 diabetes mellitus. Diabet Med. 2011; 28(11):1388–94. doi: 10. 1111/j.1464-5491.2011.03370.x PMID: 21692845.
- Zhao J, Miao K, Wang H, Ding H, Wang DW. Association between telomere length and type 2 diabetes mellitus: a meta-analysis. PLoS ONE. 2013; 8(11):e79993. doi: <u>10.1371/journal.pone.0079993</u> PMID: 24278229; PubMed Central PMCID: PMCPMC3836967.
- Bonfigli AR, Spazzafumo L, Prattichizzo F, Bonafe M, Mensa E, Micolucci L, et al. Leukocyte telomere length and mortality risk in patients with type 2 diabetes. Oncotarget. 2016. doi: 10.18632/oncotarget. 10615 PMID: 27437767.
- 55. van der Harst P, Wong LS, de Boer RA, Brouilette SW, van der Steege G, Voors AA, et al. Possible association between telomere length and renal dysfunction in patients with chronic heart failure. The American journal of cardiology. 2008; 102(2):207–10. doi: 10.1016/j.amjcard.2008.03.040 PMID: 18602523
- Melk A, Ramassar V, Helms LM, Moore R, Rayner D, Solez K, et al. Telomere shortening in kidneys with age. J Am Soc Nephrol. 2000; 11(3):444–53. PMID: 10703668.
- Zhao J, Zhu Y, Uppal K, Tran VT, Yu T, Lin J, et al. Metabolic profiles of biological aging in American Indians: the Strong Heart Family Study. Aging (Albany NY). 2014; 6(3):176–86. doi: 10.18632/aging. 100644 PMID: 24799415; PubMed Central PMCID: PMCPMC4012935.